Author's response to reviews

Title: A profile of prognostic and molecular factors in European and Maori breast cancer patients

Authors:

Gabi U Dachs (gabi.dachs@otago.ac.nz)
Maiko Kano (thpethial@gmail.com)
Ekaterina Volkova (ekaterina.volkova@otago.ac.nz)
Helen R Morrin (helen.morrin@otago.ac.nz)
Valerie CL Davey (valerie.davey@cdhb.govt.nz)
Gavin C Harris (gavin.harris@cdhb.govt.nz)
Michelle Cheale (michelle.cheale@cdhb.govt.nz)
Christopher Frampton (chris.frampton@cdhb.govt.nz)
Margaret J Currie (margaret.currie@otago.ac.nz)
J Elisabeth Wells (elisabeth.wells@otago.ac.nz)
Bridget A Robinson (bridget.robinson@cdhb.govt.nz)

Version: 3 Date: 15 September 2010

Author's response to reviews: see over
Dear Dr Rohrer-Bley

2nd Resubmission: MS 9261575303789195: ‘A profile of prognostic and molecular factors in European and Māori breast cancer patients’

Thank you for accepting our manuscript, in principal, and for the opportunity to submit a revised version. We have made the necessary changes as requested by the reviewer as outlined below:

P12, line 3: Change from Figure 1a and 1b, to Figure 1 and Figure 2.
P24: Figure legends changed to Figure 1 and Figure 2.

All changes are indicated in red in the revised manuscript.

Please do not hesitate to contact me if there are any further queries.

Yours sincerely,

Gabi Dachs